Check for updates

G OPEN ACCESS

Citation: Woodard J, Zheng W, Zhang Y (2021) Protein structural features predict responsiveness to pharmacological chaperone treatment for three lysosomal storage disorders. PLoS Comput Biol 17(9): e1009370. https://doi.org/10.1371/journal. pcbi.1009370

Editor: Piero Fariselli, Universita degli Studi di Torino, ITALY

Received: June 7, 2021

Accepted: August 21, 2021

Published: September 16, 2021

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1009370

Copyright: © 2021 Woodard et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

RESEARCH ARTICLE

Protein structural features predict responsiveness to pharmacological chaperone treatment for three lysosomal storage disorders

Jaie Woodard¹, Wei Zheng¹, Yang Zhang^{1,2}*

1 Department of Computational Medicine and Bioinformatics, University of Michigan, Ann Arbor, Michigan, United States of America, 2 Department of Biological Chemistry, University of Michigan, Ann Arbor, Michigan, United States of America

* zhng@umich.edu

Abstract

Three-dimensional structures of proteins can provide important clues into the efficacy of personalized treatment. We perform a structural analysis of variants within three inherited lysosomal storage disorders, comparing variants responsive to pharmacological chaperone treatment to those unresponsive to such treatment. We find that predicted $\Delta\Delta G$ of mutation is higher on average for variants unresponsive to treatment, in the case of datasets for both Fabry disease and Pompe disease, in line with previous findings. Using both a single decision tree and an advanced machine learning approach based on the larger Fabry dataset, we correctly predict responsiveness of three Gaucher disease variants, and we provide predictions for untested variants. Many variants are predicted to be responsive to treatment, suggesting that drug-based treatments may be effective for a number of variants in Gaucher disease. In our analysis, we observe dependence on a topological feature reporting on contact arrangements which is likely connected to the order of folding of protein residues, and we provide a potential justification for this observation based on steady-state cellular kinetics.

Author summary

Pharmacological chaperones are small molecule drugs that bind to proteins to help stabilize the folded state. One set of diseases for which this treatment has been effective is the lysosomal storage disorders, which are caused by defective lysosomal enzymes. However, not all genotypes are equally responsive to treatment. For instance, missense mutants that are particularly destabilized relative to WT are less likely to respond. The availability of datasets containing responsiveness data for large numbers of mutants, along with crystal structures of the protein involved in each disease, make machine learning methods incorporating sequence-based and structural data feasible. We hypothesize that data from two diseases, Fabry and Pompe disease, may be useful for predicting responsiveness of variants **Funding:** This work is supported (awards to YZ) in part by the National Institute of General Medical Sciences (GM136422, S100D026825), the National Institute of Allergy and Infectious Diseases (Al134678), and the National Science Foundation (IIS1901191, DBI2030790, MTM2025426). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

in the related Gaucher disease. Results suggest that many rare variants in Gaucher disease could be amenable to existing drugs. Results also suggest that drug responsiveness depends on protein topology in such a way that mutations in early-to-fold residues are more likely to be non-responsive to pharmacological chaperone treatment, which is consistent with a simple kinetic model of stability rescue. This study provides an example of how machine learning can be used to inform further studies towards personalized treatment in medicine.

Introduction

One mechanism by which a missense mutation can exert a pathogenic effect is by destabilizing the protein in which it is located, leading to deficiency of the protein, not just its enzymatic activity [1-5]. A smaller amount of the protein is then available to carry out its function, particularly since the destabilized protein is more likely to be cleared by quality control machinery [6–8]. One potential treatment for diseases caused by destabilized variants is a small molecule drug that helps stabilize the mutant protein, called a pharmacological chaperone [9–13]. Pharmacological chaperones have become popular as potential treatments for several diseases, including some lysosomal storage disorders, over the past 20 years, and the drug Migalastat (with the commercial name Galafold) based on this approach, has been approved for treatment of Fabry disease [14-16]. The drug binds the protein as an inhibitor, assisting folding in the Endoplasmic Reticulum. This follows an initial observation that the product and inhibitor galactose, of which Migalastat is an analog, improves activity and patient condition, although it is not a feasible therapeutic due to the high required dose [9,17]. Non-inhibitory chaperones have also been proposed and tested at various stages of the approval process [12,18–20], which may overcome dosing complications introduced by inhibitory effects. For lysosomal storage disorders of appropriate genotype, oral pharmacological chaperone treatment can replace or supplement intravenous enzyme replacement treatment, which is expensive and does not cross the blood-brain barrier.

The lysosomal storage disorders are a set of over 50 rare genetic diseases characterized by defective processing and build-up of substrates in the lysosome, often due to mutation in a lysosomal enzyme [21,22], which include Gaucher, Fabry, and Pompe disease. Gaucher disease [23] is the most common lysosomal storage disorder, consisting of three subtypes, the rarer two of which include neurological symptoms. Pharmacological chaperone treatment has shown promise for treatment of the disease [23,24–28], although no pharmacological chaperone drugs are currently FDA approved for this purpose.

Some mutants of a given protein are more responsive to pharmacological chaperone treatment than others. For Fabry disease, it has been shown that residues in the active site or those that grossly destabilize the protein are less likely to be responsive to treatment [16]. An interesting biophysical and bioinformatic question is then whether responsiveness to treatment can be predicted from molecular properties, utilizing machine learning methods. Previous studies have demonstrated success in predicting pathogenic variants and/or responsiveness to pharmacological chaperone treatment for Fabry disease [29–31]. Here, we seek, in particular, to elucidate how protein structure can inform upon whether a mutation is responsive to treatment. Protein structures have adopted an increasingly important role in many aspects of genetics, *e.g.*, [32–35]. We hypothesized that pathogenicity would depend with reasonably high prediction accuracy on aspects obtainable from structure and computation, such as the predicted stability change and local topology, and that predictions may be transferrable between proteins that cause differing lysosomal storage disorders.

Cell-based assays can effectively predict response to pharmacological chaperone treatment. We analyze datasets for the cases of Fabry and Pompe disease [15,36], for which large numbers of variants have been tested. We find that free energy change $\Delta\Delta G$ upon mutation predicted by our program EvoEF [37] is predictive of responsiveness to treatment, with particularly destabilized variants more likely to be unresponsive. Furthermore, we find a dependence on a topological measure which predicts that mutations in residues early to fold will be less likely to be responsive to pharmacological chaperone treatment, which we rationalize according to cellular folding kinetics. We train machine learning models on the two datasets, using a combination of sequence-based and structural features, and we use the resulting models to predict responsiveness for Gaucher disease variants. We find that many variants of Gaucher disease are predicted to be responsive to pharmacological chaperone treatment, using tested variants as a point of comparison.

Results

Key structural features of missense mutations

A number of structural features were considered for each of the three proteins associated with lysosomal storage disorder that were analyzed in this study, as illustrated in Fig 1. First, $\Delta\Delta G$ was estimated using the program EvoEF [37]. Consistent with previous results [16], mutations not responsive to pharmacological chaperone treatment showed a statistically significant increase in mean predicted $\Delta\Delta G$ of the distribution, for both Fabry and Pompe disease proteins (Fig 2). Other information we considered in decision tree analysis was the number of residues in contact with the mutated residue (6 or more heavy atom pairwise contacts within 8 Angstroms), whether the mutated residue was within 5 Angstroms of the ligand in the crystal structure, the crystallographic B-factor of the alpha carbon, and biomolecular circuit topology information based on residue-residue contacts.

Circuit topology provides a way to formalize the relationships between pairs of contacts within a protein or other linear polymer. The interval of a contact is defined as the span of



Fig 1. Illustration of structural features considered in machine learning. Features considered were the number of residues in contact with the mutated residue, whether the residue is in contact with the ligand at the catalytic site, the crystallographic B-factor or temperature factor, the change in free energy upon mutation, the "local" circuit topology of the residue, and whether the residue is located in the catalytic domain.

https://doi.org/10.1371/journal.pcbi.1009370.g001





https://doi.org/10.1371/journal.pcbi.1009370.g002

sequence between the two contacting residues, along the protein chain. Two contacts may be in three types of relations: series, parallel, or cross, as illustrated in Fig 1 and described in previous publications [38–40]. In particular, the parallel relation, relevant to this study, occurs when one interval is contained entirely within the other. A distinction is made between strict parallel and inverse parallel. One contact is said to be in parallel with another contact if it is contained in its interval, while one contact is said to be in inverse parallel with another contact if it contains its interval (i.e., the outer contact in a parallel relation). We then define local circuit topology to be the number of contacts with a particular relation relative to any contact formed by the mutated residue, consistent with (*manuscript in preparation*).

The domain architecture of each of the three proteins was considered visually in further detail (S1 Fig). Here, the central residue numbers of secondary structural elements (alpha helices and beta strands) which interact in three-dimensional space are connected by a curve. Interestingly, the non-catalytic domain in the case of glucocerebrosidase (Gaucher) consists of intertwined N- and C-terminal regions of the chain, leading to a large number of contacts in parallel. It is important to consider that differences in domain structure may affect transferability of mutational effects.

Decision tree analysis and predictions

Pharmacological chaperone responsiveness, based on enzymatic activity in cellular assays, as described in referenced publications [15,36], was predicted in this study. A decision tree based on the Fabry dataset (Fig 3A) was generated using R-part in R, using the default algorithm, with the complexity parameter determining the number of decision nodes chosen to first optimize the MCC of predictions on the Pompe dataset and then, for equivalent values on the Pompe dataset, to locally optimize the MCC of the Fabry dataset (S1 Table). The first branching is according to predicted $\Delta\Delta G$ of the mutation, where destabilization greater than 1.7 kcal/mol predicts non-responsiveness to pharmacological chaperone treatment. Next, if the mutated residue is in contact with the ligand in the crystal structure, non-responsiveness is also predicted. Otherwise, if the number of contacts, the tree again branches at $\Delta\Delta G$, where values smaller than 0.94 but greater than -1.4 kcal/mol are predicted as responsive. The most stabilizing mutations are predicted to be non-responsive. For values of $\Delta\Delta G$ greater than 0.94 (but still less than 1.7 kcal/mol), the number of inverse parallel relations determines responsiveness.

The decision tree correctly predicts responsiveness or non-responsiveness for three Gaucher disease mutations to the chemical chaperone N-nonyl-deoxynojirimycin in a cell-based assay (Table 1). Specifically, N370S and G202R are predicted to be responsive, while L444P is predicted to be non-responsive, consistent with known mutational effects [24].



Fig 3. Decision tree of structural features and importance of the inverse parallel relation. (A) Decision tree constructed in R-part from the Fabry dataset, MCC = 0.39 on Fabry dataset, MCC = 0.49 on Pompe dataset. Tree predicts whether the mutation is responsive (yes) or non-responsive (no) to pharmacological chaperone treatment. Units of $\Delta\Delta G$ are kcal/mol. (B) The orange contact is in inverse parallel (P⁻¹) relation with the gray contact. An example curve satisfying this condition (left) and topology diagram for this relation (right) are shown. (C) Model of folding of the parallel relation, with the gray contact forming first, facilitating formation of the orange contact. The top diagram shows the contacting points of the unfolded chain, with complementary points in the same color; bottom diagrams show a partially folded chain with the inner, gray contact formed (left) and a folded chain with both contacts formed (right).

https://doi.org/10.1371/journal.pcbi.1009370.g003

Table 2 shows several structural properties of these mutations. Notably, the non-responsive mutation L444P is predicted to be the most destabilizing of the first three mutations, above the 1.7 kcal/mol threshold of the decision tree. (Additional table columns are explained in later sections of the text.) The tree is less effective on predicting the responsiveness of additional variants to Ambroxol [41,42], correctly predicting only N188S (last four mutations of Table 1). Although non-responsive to N-nonyl-deoxynojirimycin, L444P was shown to be responsive in other cases [43,44].

An important aspect of the decision tree is the branching according to the number of inverse parallel relations, for intermediate $\Delta\Delta G$ values. The inverse parallel relation and implications for folding are depicted in Fig <u>3B</u> and <u>3C</u>. Considering intervals between contacting points, the contact in inverse parallel relation (orange) is outside of the first contact (gray, Fig <u>3B</u>). Formation of the gray contact will then shorten the distance along the chain of the contacting points of the orange contact (Fig <u>3C</u>). The gray contact is then expected to form first during the folding process, with the contact in inverse parallel folding later on.

To verify that residues with a large number of inverse parallel relations tend to be early to fold, we referenced hydrogen-deuterium exchange mass spectrometry experiments for a TIM barrel protein [45]. We found that residues in regions conferring the strongest protection

Та	ble 1.	Predicted	l responsiv	eness of	Gaucher	disease	mutations.
----	--------	-----------	-------------	----------	---------	---------	------------

	Observed responsive	Single tree responsive	Fabry ML responsive (p1)	Pompe ML Responsive (p1)
N370S	yes	yes (0.60)	yes (0.41)	yes (0.44)
L444P	no*	no (0.28)	no (0.24)	yes (0.42)
G202R	yes	yes (0.60)	yes (0.36)	yes (0.60)
F213I	yes	no (0.34)	yes (0.62)	no (0.30)
R120W	yes	no (0.28)	no (0.05)	no (0.24)
N188S	yes	yes (0.60)	yes (0.66)	no (0.31)
D409H	no	yes (0.60)	yes (0.38)	yes (0.42)

*Although non-responsive to N-nonyl-deoxynojirimycin, L444P was shown to be responsive in other cases [43,44].

https://doi.org/10.1371/journal.pcbi.1009370.t001

	EvoEF $\Delta\Delta G$ (kcal/mol)	Binds ligand	Number of contacts	Catalytic domain	P ⁻¹ relations
N370S	0.28	No	3	Yes	183
L444P	1.96	no	2	No	43
G202R	-1.00	No	2	Yes	219
F213I	1.05	No	5	Yes	219
R120W	2.18	No	10	Yes	145
N188S	0.42	No	2	Yes	208
D409H	0.55	No	2	Yes	79

Table 2. Structural properties of Gaucher disease mutations.

https://doi.org/10.1371/journal.pcbi.1009370.t002

from exchange (first to fold) have a higher number of inverse parallel relations on average, followed by residues exhibiting intermediate protection, followed by other residues (S2 Fig). This suggests that residues that are earlier to fold tend to have a larger number of inverse parallel relations for the TIM barrel fold, which is found in the proteins associated with lysosomal storage disorders referenced in this study.

Glycine to arginine is a severe mutation in terms of change in charge, size, and torsional propensity. It is one of the most common pathogenic mutations and has a strong bias towards pathogenicity, according to the ADDRESS database of over 20K pathogenic and benign variants [3]. However, the position of glycine 202 within a loop in the structure of glucocerebrosidase prevents the mutated residue from strongly clashing with other residues in the protein, and the predicted $\Delta\Delta G$ is in fact negative. In the ADDRESS dataset, benign glycine to arginine mutations are shifted towards more negative $\Delta\Delta G$ relative to pathogenic ones (Fig 4A), and the probability of pathogenicity for a stabilizing mutation is 0.69 vs. 0.91 for a destabilizing mutation, indicating that the small $\Delta\Delta G$ value for G202R causes the mutation to more likely be benign. Notably, G to R mutations exhibit a wide range of $\Delta\Delta G$ values, with many predicted highly destabilizing (predominantly pathogenic) mutations, as well as many predicted stabilizing (predominantly benign) ones. The EvoEF optimized structure predicts in fact that G202R is highly solvent exposed. Interestingly, two arginine residues, including the mutated residue, are in close vicinity, such that $\Delta\Delta G$ may be more positive than predicted.

Kinetic model of folding in the Endoplasmic Reticulum

To establish whether the dependence of drug responsiveness on the number of inverse parallel relations could be explained by kinetics of the folding and interaction system, we constructed a simple kinetic model. Referencing Fig 3B and 3C, residues that fold early in the folding process are expected to have a large number of contacts in inverse parallel relation. According to the two-state folding model [46,47], mutation of such early folding residues is expected to change



Fig 4. Glycine to arginine mutation statistics and predicted structure. (A) Histograms of $\Delta\Delta G$ (kcal/mol) from the ADDRESS database for glycine to arginine mutations. T-test p-value = $2x10^{-10}$. (B) EvoEF optimized structure for the G202R mutation (using the BuildMutant command within EvoEF).

https://doi.org/10.1371/journal.pcbi.1009370.g004

the folding rate, rather than the unfolding rate. Intuitively, we might expect that mutations that speed the unfolding rate (*i.e.*, with few inverse parallel relations) could be more easily rescuable by stabilizing molecules that bind to the folded state. This would be consistent with the results of the decision tree in Fig 3, which predicts responsiveness for few numbers of parallel relations but non-responsiveness for large numbers of parallel relations.

A kinetic model quantifying this argument is shown in Fig 5. Unfolded protein, U, is produced on polyribosomes and enters the Endoplasmic Reticulum (ER). The protein folds to state F with rate k_{fr} or is degraded at rate k_{pr} . From the folded state, the protein may unfold back to state U with rate k_u . It may also bind the pharmacological chaperone ligand with rate konL, or it may be exported from the ER with rate kout to state E. From the chaperone-bound state F_B , the protein may dissociate from the chaperone and re-enter state F, or it may also be exported to state E. Steady-state results for parameter values given in the methods section, according to Eqs 1–3, are shown in Fig 5B, for mutations destabilizing the protein by speeding the unfolding rate (blue, cyan) or slowing the folding rate (red, magenta). Mutations that speed the unfolding rate are seen to be more responsive to treatment for a given $\Delta\Delta G$ value. While in actuality, mutations are likely to have fractional phi values, affecting both folding and unfolding rates to some extent, and the protein is likely to have multi-state folding properties, this model suggests that the dependence on number of inverse parallel relations may be explainable by protein folding propensities. While it is possible that additional biophysical factors contribute to the trend of circuit-topology-dependent responsiveness, it is interesting that both a kinetic model based on a reasonable model of enzyme supply to the lysosome and the observation that large numbers of inverse parallel relations decrease the chance of responsiveness are consistent with a picture in which speeding of the unfolding process is more likely to be rescued by pharmacological chaperone treatment than slowing of the folding process, for equivalent $\Delta\Delta G$.

$$F = \frac{k_f P}{k_p + k_f} \left(k_u + k_{on} + k_{out} - \frac{k_f k_u}{k_p + k_f} - \frac{k_{off} k_{on}}{k_{off} + k_{out}} \right)$$
(1)

$$F_B = \frac{k_{on}F}{k_{off} + k_{out}} \tag{2}$$

$$\frac{dE}{dt} = k_{out}(F + F_B) \tag{3}$$



Fig 5. Kinetic model of ER folding and export, with and without pharmacological chaperone (PC). (A) The kinetic model, including unfolded state, U, folded state F, folded state bound to chaperone F_{B} , and protein exported from the ER E. P is the rate of production of unfolded protein, k_f is the folding rate, k_u is the unfolding rate, $k_{on}L$ is the rate of chaperone binding, k_{off} is the rate of dissociation of protein and chaperone, and k_{out} is the rate of export. (B) Steady state solutions for rate of export, $\frac{dt}{dt}$, for mutations affecting the folding or unfolding rate, with or without pharmacological chaperone (PC) treatment.

https://doi.org/10.1371/journal.pcbi.1009370.g005

Machine learning of responsiveness to pharmacological chaperone treatment

Machine learning was carried out using the Auto-ML capability of h2o. For the Fabry dataset, the best performing model was an ensemble of methods, while for the Pompe dataset it was an XGBoost method (S2 Table). Cross-validation AUC was high in both cases: 0.829 and 0.881, respectively (Table 3). The MCC on the same dataset (assessing performance on the cross-validated data of the entire collection of mutations) was 0.839 for Fabry and 0.712 for Pompe. In addition to cross validation on the same dataset, we obtained the MCC using the other dataset as the test set. (i.e., training on the Fabry dataset and testing non the Pompe dataset, or training on the Pompe dataset and testing on the Fabry dataset). The MCC for the performance of the Fabry-trained model on the Pompe dataset was 0.609, while the MCC of the Pompe-trained dataset on the Fabry data is substantially lower (0.282), likely because of a smaller amount of training data that does not encompass the full feature space of the Fabry dataset.

We predicted drug responsiveness for Gaucher disease variants based on the other two datasets. 105 of 203 mutations were predicted by both methods to be responsive to treatment, and 47 were predicted as non-responsive by both methods, with an MCC comparing the predictions of the two methods on the Gaucher dataset of 0.455. The prevalent mutation N370S was predicted to be responsive to treatment by both the Fabry and Pompe dataset based methods, as was G202R. L444P, which is known to be more difficult to treat via pharmacological chaperones [24], was predicted as non-responsive by the Fabry based method and responsive by the Pompe based method (although it had the lowest p1 of the three mutations, see Table 1). Complete predictions are given in S3 Table.

In general, based on our statistics and properties of the training data and models, we might expect the machine learning (H2O) results based on the Fabry dataset to yield the highest accuracy results. We in fact see better performance of the Fabry machine learning model on the additional Ambroxol responsiveness variants (last four mutations, Table 1). Of the two mutations incorrectly predicted, D409H had a relatively low p1 value. R120W, however, was shown experimentally to be responsive to Ambroxol despite a very low p1. Table 2 shows that this variant has several properties predicted to increase the propensity to be non-responsive to treatment, including a $\Delta\Delta G$ of greater than 1.7 kcal/mol and a large number of contacts involving the mutated residue. It could be informative to study this mutant in further detail using computational and/or experimental methods. The larger dataset of Fabry disease, combined with the sophisticated machine learning method, yields increased performance over the other methods described.

A molecular dynamics study [29] reported that glycine residues in particular had lower residual activity than might be expected based on RMSF, in relation to other residues. Glycine in general may be expected to have high flexibility, due to the large amount of torsional accessibility in the Ramachandran map relative to other residues; in addition, referencing our work summarizing mutations in 70,597 proteins [3], and as also seen previously [48], mutations from glycine to other residues tend to be pathogenic more often than mutations from most other residues (see <u>\$3 Fig</u>), which could be a reason for the observed anomalous behavior of glycine mutations.

Table 3. Results of cross validation incorporating H2O Auto-ML.

Dataset	AUC same	MCC same	MCC other
Fabry	0.829	0.734	0.609
Pompe	0.881	0.712	0.282

https://doi.org/10.1371/journal.pcbi.1009370.t003

Discussion

Pharmacological chaperone treatment has been explored in the case of several diseases [9,10,21], with much of the focus centering on certain lysosomal storage disorders. For such disorders, the small molecule chaperone helps to stabilize destabilized but still potentially enzymatically active variants of an enzyme. In drug-amenable proteins, key residues for catalytic activity are intact, while the variants affect the stability of folding. Prediction of which variants are amenable to such treatment has similarities to the prediction of pathogenicity, and factors such as excessive destabilization and ligand-contact, which contribute towards pathogenicity, tend also to contribute towards lack of rescue. However, the range of these values across variants is intrinsically shortened in the case of chaperone responsiveness, since all mutations are pathogenic.

It was suggested that the L444P variant was non-responsive to treatment due to its location in a domain other than the one containing the active site which is bound by the drug [24]. However, referencing the Fabry dataset [15], we see that mutations in residues not contained in the active site domain are often responsive to pharmacological chaperone treatment. The leucine to proline mutation type generally has a strong bias towards pathogenicity [3]. Therefore, the mutation type and stability change of the L444P mutation may be the source of pathogenicity, rather than (or in addition to) domain location. Considering these factors, it may be useful to test for responsiveness of other mutants in the non-catalytic domain of glucocerebrosidase. For instance, R463C, which comes in contact with the catalytic domain, is predicted to be responsive to treatment.

We note that our machine learning approach does not explicitly account for amino acid change propensities in ADDRESS [3]; these may be considered in addition to the given predictions. For instance, although arginine to cysteine mutations are most often pathogenic, these mutations are common and are benign in many cases. In fact, in the case of Fabry disease, several arginine to cysteine mutations show residual basal activity and responsiveness to pharmacological chaperone treatment [13]. Since similar propensities that cause a mutation to be pathogenic also cause it to be non-responsive to treatment (e.g., strong destabilization), a transfer learning approach, utilizing data on pathogenicity may be applicable.

It is interesting that the G202R mutation, which is known to be rescuable by pharmacological chaperone treatment, is predicted to stabilize, rather than destabilize the protein. Referencing Fig 2, this is the case for many variants. It is possible that some variants detrimentally affect folding without increasing the locally computed $\Delta\Delta G$. For instance, a mutation may make the protein more prone to misfold upon global unfolding. The bound drug would decrease the propensity for unfolding, reducing the propensity for misfolding and degradation. Kinetics may also be important [6,49]. A mutation that approximately equally slows unfolding and slows folding may have a pathogenic effect, for instance, if there is a particular importance of folding (vs. unfolding) rate. Although variants that are predicted to be especially stabilized are more likely to be non-responsive to treatment, the case of enzymes with $\Delta\Delta G$ predicted as sufficiently stabilizing to drive the prediction of non-responsiveness is fairly rare across clinically observed disease-causing mutations.

Our results support the key biological insight that kinetics of folding and binding in the cellular environment plays an important role in determining the responsiveness of variants to pharmacological chaperone treatment in the case of lysosomal storage disorders. Kinetic information is contained in circuit topology features derived from the crystal structure; contacts with many inverse parallel relations are assumed to be early to fold. According to a kinetic model built to approximate folding and drug binding in the cell, early to fold contacts are more difficult to rescue. This is consistent with the observation that mutants of residues with a large number of inverse parallel relations are statistically less likely to be responsive to pharmacological chaperone treatment. These factors link cellular kinetics and protein topology and demonstrate that both are likely important in determining drug responsiveness in this system. It will be important to assess how results of molecular dynamics [29,50] and monte carlo simulations (e.g., RMSF or number of native contacts at point of residue unfolding) correlate with residue-based circuit topology. In the case that they are somewhat redundant, circuit topology measures might be a quick alternative to more computationally intensive simulations, suitable for analysis of large numbers of proteins at once, and in the case that they provide unique information, they might serve complementary roles in structure-based analysis. Our results and models suggest that different systems might depend differently on the numbers of relations of a given type, depending on details of the kinetics of the system.

It will be interesting to investigate the transferability of information from these lysosomal storage disorders, along with larger amounts of information on pathogenicity of human variants, to the proteins involved in other pharmacological-chaperone-amenable diseases. Predictions on responsiveness of variants can be used when considering whether a treatment may be effective, as well as for gauging the difficulty of the target when pursuing drug design. While we have utilized crystallographic structures in this study, NMR structures and computationally predicted structures and ligand binding sites may also be utilized in future studies of other proteins. Ultimately, protein-centric methods can be extended to other disease types, including cancers, with the aim of improving precision medicine based treatment decisions and facilitating drug discovery.

Methods

Datasets and structures

Existing datasets were referenced for response to pharmacological chaperones for Fabry [15] and Pompe [36] diseases. A list of Gaucher disease variants was obtained from a published article [23]. Crystal structures 2V3D (Gaucher [51]), 3GXT (Fabry [52]), and 5NN5 (Pompe [53]) were referenced.

Domain divisions

For 2V3D and 3GXT, Pfam was referenced to determine domain divisions, and the catalytic domain was identified by visual inspection. For 5NN5, Pfam incorrectly determined domain divisions. This structure was aligned to the first two structures using TM-align [54], and domain divisions were determined according to the alignment. The ending residue of the catalytic domain was different for alignment to the two structures, and the average sequence position was chosen as the ending point.

Protein free energy change upon mutation

Change in free energy upon mutation, $\Delta\Delta G$, was estimated using the program EvoEF [37]. The mutant protein was built and refined in EvoEF, following refinement of the WT protein and subsequent calculation of reference free energy for comparison.

Contact information

Number of residues in contact with the mutated residue was calculated for each residue type by considering the number of residues forming greater than or equal to five atom-atom pairwise contacts of any type within 4.5 Angstroms. Circuit topology relations were calculated using existing code and methods [38]. For each residue, the number of relations of a given type with respect to any contact formed by the mutated residue was considered as "local" circuit topology for that residue.

PSSM generation

To construct the PSSM, multiple sequence alignments (MSAs) for the query sequence were first generated by DeepMSA2, which utilizes HHblits [55,56], Jackhmmer [57] and HMMsearch [58] to iteratively search two whole-genome sequence databases (Uniclust30 [59] and UniRef90 [60]) and four metagenome sequence databases (Metaclust [61], BFD [62], Mgnify [63], and IMG/M [64]). DeepMSA2 contains three approaches, dMSA, qMSA, and mMSA, where the dMSA pipeline is equivalent to the DeepMSA [65] pipeline.

For the dMSA pipeline, HHblits2, Jackhmmer and HMMsearch were used to search the query against Uniclust30 (version 2017_04), UniRef90 and Metaclust, respectively. In Stage 2 and 3 of dMSA, homologs identified by Jackhmmer and HMMsearch, respectively, were gathered into a custom HHblits format database, which would be searched by HHblits2 using the MSA input from the previous stage to generate new MSAs. As an extension of dMSA, qMSA (which stands for "quadruple MSA") performs HHblits2, Jackhmmer, HHblits3, and HMMsearch searches against Uniclust30 (version 2020 01), UniRef90, BFD, and Mgnify, respectively, in four stages. Similar to dMSA Stage 2 and 3, the sequence hits from Jackhmmer, HHblits3 and HMMsearch in Stage 2, 3 and 4 of qMSA were converted into HHblits format database, against which the HHblits2 search based on MSA input from the previous stage is performed. In mMSA (or "multi-level MSA"), the qMSA Stage 3 alignment was used as a probe by HMMsearch to search through the IMG/M database and the resulting sequence hits were converted into a sequence database. This mMSA database was then used as the target database, which was searched by HHblits2 with three seed MSAs (MSA from dMSA stage 2, qMSA stage 2 and 3), to derive three new MSAs. These steps resulted in 10 MSAs in total (i.e., 3 from dMSA, 4 from qMSA, and 3 from mMSA), which were scored by TripletRes [66] contact prediction, where the MSA with the highest probabilities for top 10L (L is the sequence length) all range contacts (C β -C β distances<8Å) were be selected.

The final MSA from DeepMSA2 is formatted as "a3m" file. This "a3m" format file was first converted to Blast [67] format database. Then, the query sequence was used to search this database, using PSI-Blast, and finally the PSSM file was generated by PSI-Blast.

Kinetic model

Equations for the kinetic model are given in the main text. Kinetic parameters were chosen as: P = 0.1, $k_p = 10$, $k_{on}L = 0.1$ (or 0.00001 without chaperone), $k_{off} = 0.01$ (or 0.00001 without chaperone), $k_{out} = 0.01$, $k_u = 0.01$ or ranging from 0.005 to 10 in increments of 0.0005, $k_f = 10$ or ranging from 0.01 to 20 in increments of 0.001, $\Delta\Delta G = 4.0963 - 0.593^* \ln(k_f/k_u)$.

Machine learning

The machine learning feature set consisted of the type of residues mutated from and to (employing one-hot encoding), the number of residues of each type in contact with the mutated residue, the crystallographic B-factor of the alpha carbon, the EvoEF-predicted $\Delta\Delta G$ of mutation, whether each residue was within 5 Angstroms of the active site ligand, and the deep-Multiple-Sequence-Alignment PSSM data for a given residue.

AutoML was performed using the program h2o, with 10-fold cross validation, using default options, and a maximum runtime of 500 seconds.

Supporting information

S1 Fig. Diagrams of proteins with mutants associated with lysosomal storage disorders. (PDF)

S2 Fig. Number of inverse parallel relations, by protection from exchange in mass spectrometry experiments.
(PDF)
S3 Fig. Propensity for benign mutation by amino acid type.
(PDF)
S1 Table. Optimization of Fabry decision tree complexity.
(PDF)
S2 Table. Leading auto-ML models.
(PDF)
S3 Table. List of complete predictions on 203 mutations.

(XLSX)

Acknowledgments

The authors thank Cameron Fen for help with Auto-ML, Eric Bell for helpful discussions, and Dr. Gil Omenn for comments on the manuscript.

Author Contributions

Conceptualization: Jaie Woodard, Yang Zhang.

Funding acquisition: Yang Zhang.

Investigation: Jaie Woodard, Wei Zheng.

Methodology: Jaie Woodard, Wei Zheng.

Supervision: Yang Zhang.

Validation: Wei Zheng.

Writing – original draft: Jaie Woodard.

Writing - review & editing: Jaie Woodard, Wei Zheng, Yang Zhang.

References

- 1. Waters P J. Degradation of mutant proteins, underlying "loss of function" phenotypes, plays a major role in genetic disease. Curr Issues Mol Biol 3, 57–65 (2001). PMID: <u>11488412</u>
- Gerasimavicius L, Liu X, Marsh JA. Identification of pathogenic missense mutations using protein stability predictors. Sci Rep 10, 15387 (2020). https://doi.org/10.1038/s41598-020-72404-w PMID: 32958805
- Woodard JC. Zhang C, Zhang Y. ADDRESS: A Database of Disease-associated Human Variants Incorporating Protein Structure and Folding Stabilities. Journal of molecular biology, 166840 (2021). https://doi.org/10.1016/j.jmb.2021.166840 PMID: 33539887
- Sanchez-Ruiz JM, Protein kinetic stability. Biophys Chem 148, 1–15 (2010). https://doi.org/10.1016/j. bpc.2010.02.004 PMID: 20199841
- Stein A, Fowler DM, Hartmann-Petersen R, Lindorff-Larsen K. Biophysical and Mechanistic Models for Disease-Causing Protein Variants. Trends Biochem Sci 44, 575–588 (2019). https://doi.org/10.1016/j. tibs.2019.01.003 PMID: 30712981
- Bershtein S, Mu W, Serohijos AW, Zhou J, Shakhnovich EI. Protein quality control acts on folding intermediates to shape the effects of mutations on organismal fitness. Mol Cell 49, 133–144 (2013). https://doi.org/10.1016/j.molcel.2012.11.004 PMID: 23219534
- Needham PG, Guerriero PG, Brodsky JL. Chaperoning Endoplasmic Reticulum-Associated Degradation (ERAD) and Protein Conformational Diseases. Cold Spring Harb Perspect Biol 11, a033928 (2019). https://doi.org/10.1101/cshperspect.a033928 PMID: 30670468

- Rutishauser J, Spiess M. Endoplasmic reticulum storage diseases. Swiss Med Wkly 132, 211–222 (2002). PMID: 12087487
- Liguori L, Monticelli M, Allocca M, Mele BH, Lukas J, Cubellis MV, et al. Pharmacological Chaperones: A Therapeutic Approach for Diseases Caused by Destabilizing Missense Mutations. Int J Mol Sci 21, 489 (2020). https://doi.org/10.3390/ijms21020489 PMID: 31940970
- Leidenheimer NJ. Pharmacological Chaperones: Beyond Conformational Disorders. Handb Exp Pharmacol 245, 135–153 (2018). https://doi.org/10.1007/164_2017_68 PMID: 29071508
- Aubi O, Knappskog PM, Martinez A. Early Stage Discovery and Validation of Pharmacological Chaperones for the Correction of Protein Misfolding Diseases. Methods Mol Biol 1873, 279–292 (2019). https://doi.org/10.1007/978-1-4939-8820-4_18 PMID: 30341617
- Tran ML, Genisson Y, Ballereau S, Dehoux C. Second-Generation Pharmacological Chaperones: Beyond Inhibitors. Molecules 25, 3145 (2020). https://doi.org/10.3390/molecules25143145 PMID: 32660097
- Citro V, Cammisa M, Ligori L, Cimmarute C, Lukas J, Cubellis MV, et al. The Large Phenotypic Spectrum of Fabry Disease Requires Graduated Diagnosis and Personalized Therapy: A Meta-Analysis Can Help to Differentiate Missense Mutations. Int J Mol Sci 17, 2010 (2016). <u>https://doi.org/10.3390/ijms17122010 PMID: 27916943</u>
- Kok K, Swiers KC, Boot RG, Overkleeft HS, Aerts JMFG, Artola M, et al. Fabry Disease: Molecular Basis, Pathophysiology, Diagnostics and Potential Therapeutic Directions. Biomolecules 11, 271 (2021). https://doi.org/10.3390/biom11020271 PMID: 33673160
- Benjamin ER, Valle MCD, Wu X, Katz E, Pruthi F, Bond S, et al. The validation of pharmacogenetics for the identification of Fabry patients to be treated with migalastat. Genet Med 19, 430–438 (2017). https://doi.org/10.1038/gim.2016.122 PMID: 27657681
- Andreotti G, Guarracino MR, Cammisa M, Correra A, Cubellis MV, Prediction of the responsiveness to pharmacological chaperones: lysosomal human alpha-galactosidase, a case of study. Orphanet J Rare Dis 5, 36 (2010). https://doi.org/10.1186/1750-1172-5-36 PMID: 21138548
- Okumiya T, Ishii S, Takenaka T, Kase R, Kamei S, Sakuraba H, et al. Galactose stabilizes various missense mutants of alpha-galactosidase in Fabry disease. Biochem Biophys Res Commun 214, 1219–1224 (1995). https://doi.org/10.1006/bbrc.1995.2416 PMID: 7575533
- Porto C, Ferrara MC, Meli M, Acampora E, Avolio V, Rosa M, et al. Pharmacological enhancement of αglucosidase by the allosteric chaperone N-acetylcysteine. Mol Ther 20, 2201–2211 (2012). https://doi. org/10.1038/mt.2012.152 PMID: 22990675
- Motabar O, Sidransky E, Goldin E, Zheng W, Fabry disease—current treatment and new drug development. Curr Chem Genomics 4, 50–56 (2010). https://doi.org/10.2174/1875397301004010050 PMID: 21127742
- Jung O, Patnaik S, Marugan J, Sidransky E, Westbroek W. Progress and potential of non-inhibitory small molecule chaperones for the treatment of Gaucher disease and its implications for Parkinson disease. Expert Rev Proteomics 13, 471–479 (2016). <u>https://doi.org/10.1080/14789450.2016.1174583</u> PMID: 27098312
- 21. Pinto EVF, Málaga DR, Kubaski F, de Souza CFM, Poswar FO, Baldo G, et al. Precision Medicine for Lysosomal Disorders. Biomolecules 10, 1110 (2020).
- Parenti G,Medina DL, Ballabio A. The rapidly evolving view of lysosomal storage diseases. EMBO Mol Med 13, e12836 (2021). https://doi.org/10.15252/emmm.202012836 PMID: 33459519
- Hruska KS, LaMarca ME, Scott CR, Sidransky E. Gaucher disease: mutation and polymorphism spectrum in the glucocerebrosidase gene (GBA). Hum Mutat 29, 567–583 (2008). <u>https://doi.org/10.1002/ humu.20676 PMID: 18338393</u>
- Sawkar AR, Adamski-Werner SL, Cheng W, Wong C, Beutler E, Zimmer K, et al. Gaucher diseaseassociated glucocerebrosidases show mutation-dependent chemical chaperoning profiles. Chem Biol 12, 1235–1244 (2005). https://doi.org/10.1016/j.chembiol.2005.09.007 PMID: 16298303
- Alfonso P, Pampin S, Estrada J, Rodríguez-Rey JC, Giraldo P, Sancho J, et al. Miglustat (NB-DNJ) works as a chaperone for mutated acid beta-glucosidase in cells transfected with several Gaucher disease mutations. Blood Cells Mol Dis 35, 268–276 (2005). https://doi.org/10.1016/j.bcmd.2005.05.007 PMID: 16039881
- Narita A, Shirai K, Itamura S, Matsuda A, Ishihara A, Matsushita K, et al. Ambroxol chaperone therapy for neuronopathic Gaucher disease: A pilot study. Ann Clin Transl Neurol 3, 200–215 (2016). https:// doi.org/10.1002/acn3.292 PMID: 27042680
- Maegawa GH, Tropak MB, Buttner JD, Rigat BA, Fuller MF, Pandit D, et al. Identification and characterization of ambroxol as an enzyme enhancement agent for Gaucher disease. J Biol Chem 284, 23502– 23516 (2009). https://doi.org/10.1074/jbc.M109.012393 PMID: 19578116

- Kim YM, Yum M, Heo SH, Kim T, Jin HK, Bae J, et al. Pharmacologic properties of high-dose ambroxol in four patients with Gaucher disease and myoclonic epilepsy. J Med Genet 57, 124–131 (2020). https://doi.org/10.1136/jmedgenet-2019-106132 PMID: 31649052
- Cubellis MV, Baaden M, Andreotti G. Taming molecular flexibility to tackle rare diseases. Biochimie 113, 54–58 (2015). https://doi.org/10.1016/j.biochi.2015.03.018 PMID: 25841341
- Saito S, Ohno K, Sese J, Sugawara K, Sakuraba H. Prediction of the clinical phenotype of Fabry disease based on protein sequential and structural information. J Hum Genet 55, 175–178 (2010). https:// doi.org/10.1038/jhg.2010.5 PMID: 20139917
- Riera C, Lois S, Domínguez C, Fernandez-Cadenas I, Montaner J, Rodríguez-Sureda V, et al. Molecular damage in Fabry disease: characterization and prediction of alpha-galactosidase A pathological mutations. Proteins 83, 91–104 (2015). https://doi.org/10.1002/prot.24708 PMID: 25382311
- Laskowski RA, Thornton JM. Understanding the molecular machinery of genetics through 3D structures. Nat Rev Genet 9, 141–151 (2008). https://doi.org/10.1038/nrg2273 PMID: 18160966
- Fanelli F, Felline A, Marigo V. Structural aspects of rod opsin and their implication in genetic diseases. Pflug Arch Eur J Phy, in press (2021). https://doi.org/10.1007/s00424-021-02546-x PMID: 33728518
- Liberles DA, Teichmann SA, Bahar I, Bastolla U, Bloom J, Bornberg-Bauer E, et al. The interface of protein structure, protein biophysics, and molecular evolution. Protein Sci 21, 769–785 (2012). <u>https://doi.org/10.1002/pro.2071</u> PMID: 22528593
- Alibés A, Nadra AD, Masi FD, Bulyk ML, Serrano L, Stricher F. Using protein design algorithms to understand the molecular basis of disease caused by protein-DNA interactions: the Pax6 example. Nucleic Acids Res 38, 7422–7431 (2010). https://doi.org/10.1093/nar/gkq683 PMID: 20685816
- Flanagan JJ, Rossi B, Tang K, Wu X, Mascioli, Donaudy F, et al. The pharmacological chaperone 1deoxynojirimycin increases the activity and lysosomal trafficking of multiple mutant forms of acid alpha-glucosidase. Hum Mutat 30, 1683–1692 (2009). <u>https://doi.org/10.1002/humu.21121</u> PMID: 19862843
- Pearce R, Huang X, Setiawan D, Zhang Y. EvoDesign: Designing Protein-Protein Binding Interactions Using Evolutionary Interface Profiles in Conjunction with an Optimized Physical Energy Function. Journal of molecular biology 431, 2467–2476 (2019). https://doi.org/10.1016/j.jmb.2019.02.028 PMID: 30851277
- **38.** Schullian O, Woodard J, Tirandaz A, Mashaghi A. A Circuit Topology Approach to Categorizing Changes in Biomolecular Structure. Front Phys-Lausanne 8, 5 (2020).
- Scalvini B, Sheikhhassani V, Woodard J, Dame RT, Jerala R, Mashaghi A. Topology of Folded Molecular Chains: From Single Biomolecules to Engineered Origami. Trends in Chemistry 2, 609–622 (2020).
- Mashaghi A, van Wijk RJ, Tans SJ. Circuit topology of proteins and nucleic acids. Structure 22, 1227– 1237 (2014). https://doi.org/10.1016/j.str.2014.06.015 PMID: 25126961
- Luan Z, Li L, Higaki K, Nanba E, Suzuki Y, Ohno K. The chaperone activity and toxicity of ambroxol on Gaucher cells and normal mice. Brain Dev 35, 317–322 (2013). <u>https://doi.org/10.1016/j.braindev.</u> 2012.05.008 PMID: 22682976
- Lei K, Ninomiya H, Suzuki M, Inoue T, Sawa M, Iida M, et al. Enzyme enhancement activity of N-octylbeta-valienamine on beta-glucosidase mutants associated with Gaucher disease. Biochim Biophys Acta 1772, 587–596 (2007). https://doi.org/10.1016/j.bbadis.2007.02.003 PMID: 17363227
- 43. García-Moreno MI, de la Mata M, Sánchez-Fernández EM, Benito JM, Díaz-Quintana A, Fustero S, et al. Fluorinated Chaperone-β-Cyclodextrin Formulations for β-Glucocerebrosidase Activity Enhancement in Neuronopathic Gaucher Disease. J Med Chem 60, 1829–1842 (2017). <u>https://doi.org/10.1021/acs.jmedchem.6b01550</u> PMID: 28171725
- 44. de la Mata M, Cotán D, Oropesa- Ávila M, Garrido-Maraver J, Cordero M, Paz MV, et al. Pharmacological Chaperones and Coenzyme Q10 Treatment Improves Mutant β-Glucocerebrosidase Activity and Mitochondrial Function in Neuronopathic Forms of Gaucher Disease. Sci Rep 5, 10903 (2015). https:// doi.org/10.1038/srep10903 PMID: 26045184
- 45. Jain R, Muneeruddin K, Anderson J, Harms MJ, Shaffer SA, Matthews CR. A conserved folding nucleus sculpts the free energy landscape of bacterial and archaeal orthologs from a divergent TIM barrel family. Proceedings of the National Academy of Sciences of the United States of America 118, e2019571118 (2021). https://doi.org/10.1073/pnas.2019571118 PMID: 33875592
- 46. Tian J, Woodard JC, Whitney A, Shakhnovich EI, Thermal stabilization of dihydrofolate reductase using monte carlo unfolding simulations and its functional consequences. PLoS Comput Biol 11, e1004207 (2015). https://doi.org/10.1371/journal.pcbi.1004207 PMID: 25905910
- 47. Fersht AR, Sato S, Phi-value analysis and the nature of protein-folding transition states. Proc Natl Acad Sci U S A 101, 7976–7981 (2004). https://doi.org/10.1073/pnas.0402684101 PMID: 15150406

- Gao M, Zhou H, Skolnick J. Insights into Disease-Associated Mutations in the Human Proteome through Protein Structural Analysis. Structure 23, 1362–1369 (2015). https://doi.org/10.1016/j.str. 2015.03.028 PMID: 26027735
- 49. Godoy-Ruiz R, Ariza F, Rodriguez-Larrea D, Perez-Jimenez R, Ibarra-Molero B, Sanchez-Ruiz JM, et al., Natural selection for kinetic stability is a likely origin of correlations between mutational effects on protein energetics and frequencies of amino acid occurrences in sequence alignments. J Mol Biol 362, 966–978 (2006). https://doi.org/10.1016/j.jmb.2006.07.065 PMID: 16935299
- 50. Ponzoni L, Bahar, I Structural dynamics is a determinant of the functional significance of missense variants. Proc Natl Acad Sci U S A 115, 4164–4169 (2018). <u>https://doi.org/10.1073/pnas.1715896115</u> PMID: 29610305
- Brumshtein B, Greenblatt HM, Butters TD, Shaaltiel Y, Aviezer D, Silman I, et al. Crystal structures of complexes of N-butyl- and N-nonyl-deoxynojirimycin bound to acid beta-glucosidase: insights into the mechanism of chemical chaperone action in Gaucher disease. J Biol Chem 282, 29052–29058 (2007). https://doi.org/10.1074/jbc.M705005200 PMID: 17666401
- 52. Lieberman RL, D'aquino JA, Ringe D, Petsko GA. Effects of pH and iminosugar pharmacological chaperones on lysosomal glycosidase structure and stability. Biochemistry 48, 4816–4827 (2009). <u>https://doi.org/10.1021/bi9002265 PMID: 19374450</u>
- Roig-Zamboni V, Cobucci-Ponzano B, Iacono R, Ferrara MC, Germany S, Bourne Y, et al. Structure of human lysosomal acid α-glucosidase-a guide for the treatment of Pompe disease. Nat Commun 8, 1111 (2017). https://doi.org/10.1038/s41467-017-01263-3 PMID: 29061980
- Zhang Y, Skolnick J, TM-align: a protein structure alignment algorithm based on the TM-score. Nucleic acids research 33, 2302–2309 (2005). https://doi.org/10.1093/nar/gki524 PMID: 15849316
- Steinegger M, Meier M, Mirdita M, Vöhringer H, Haunsberger SJ, Söding J. HH-suite3 for fast remote homology detection and deep protein annotation. BMC Bioinformatics 20, 473 (2019). https://doi.org/ 10.1186/s12859-019-3019-7 PMID: 31521110
- Remmert M, Biegert A, Hauser A, Söding J. HHblits: lightning-fast iterative protein sequence searching by HMM-HMM alignment. Nat Methods 9, 173–175 (2011). <u>https://doi.org/10.1038/nmeth.1818</u> PMID: 22198341
- Johnson LS, Eddy SR, Portugaly E. Hidden Markov model speed heuristic and iterative HMM search procedure. BMC Bioinformatics 11, 431 (2010). https://doi.org/10.1186/1471-2105-11-431 PMID: 20718988
- Potter CS, Luciani A, Eddy SF, Park Y, Lopez R, Finn RD. HMMER web server: 2018 update. Nucleic Acids Res 46, W200–W204 (2018). https://doi.org/10.1093/nar/gky448 PMID: 29905871
- Mirdita M, von den Driesch L, Galiez C, Martin MJ, Söding J, Steineger M. Uniclust databases of clustered and deeply annotated protein sequences and alignments. Nucleic Acids Res 45, D170–D176 (2017). https://doi.org/10.1093/nar/gkw1081 PMID: 27899574
- Consortium U, UniProt: a worldwide hub of protein knowledge. Nucleic Acids Res 47, D506–D515 (2019). https://doi.org/10.1093/nar/gky1049 PMID: 30395287
- Steinegger M, Soding J. Clustering huge protein sequence sets in linear time. Nat Commun 9, 2542 (2018). https://doi.org/10.1038/s41467-018-04964-5 PMID: 29959318
- Steinegger M, Mirdita M, Söding J. Protein-level assembly increases protein sequence recovery from metagenomic samples manyfold. Nat Methods 16, 603–606 (2019). https://doi.org/10.1038/s41592-019-0437-4 PMID: 31235882
- Mitchell AL, Almeida A, Beracochea M, Boland M, Burgin J, Cochrane G, et al. MGnify: the microbiome analysis resource in 2020. Nucleic Acids Res 48, D570–D578 (2020). https://doi.org/10.1093/nar/ gkz1035 PMID: 31696235
- Chen IA, Chu K, Palaniappan K, Pillay M, Ratner A. Huang J, et al. IMG/M v.5.0: an integrated data management and comparative analysis system for microbial genomes and microbiomes. Nucleic Acids Res 47, D666–D677 (2019). https://doi.org/10.1093/nar/gky901 PMID: 30289528
- Zhang C, Zheng W, Mortuza SM, Li Y, Zhang Y. DeepMSA: constructing deep multiple sequence alignment to improve contact prediction and fold-recognition for distant-homology proteins. Bioinformatics 36, 2105–2112 (2020). https://doi.org/10.1093/bioinformatics/btz863 PMID: 31738385
- Li Y, Zhang C, Bell EW, Yu JD, Zhang Y, Ensembling multiple raw coevolutionary features with deep residual neural networks for contact-map prediction in CASP13. Proteins 87, 1082–1091 (2019). https://doi.org/10.1002/prot.25798 PMID: 31407406
- Altschul SF, Gish W, Miller W, Myers EW, Lipman DJ, Basic local alignment search tool. Journal of molecular biology 215, 403–410 (1990). https://doi.org/10.1016/S0022-2836(05)80360-2 PMID: 2231712